Author's response to reviews

Title: Psychological adjustment and quality of life in children and adolescents following open-heart surgery for congenital heart disease: a systematic review

Authors:

Beatrice Latal (bea.latal@kispi.uzh.ch)
Susanne Helfricht (susanne.helfricht@gmx.net)
Joachim E Fischer (jfischer@medma.uni-heidelberg.de)
Urs Bauersfeld (urs.bauersfeld@kispi.uzh.ch)
Markus A Landolt (markus.landolt@kispi.uzh.ch)

Version: 5 Date: 19 January 2009

Author's response to reviews: see over
Dear Editor

We thank you for the acceptance of our manuscript. We have made the requested formatting changes according to your instructions. In particular, we have transformed the Tables 2a and 2b into attached files 1 and 2.

With best regards,

Yours sincerely

Markus A. Landolt, Ph.D.
Corresponding author